\documentclass[11pt]{article}
%\usepackage[margin=1in]{geometry}
\usepackage{amsfonts,amsmath,amsthm,array,amssymb}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}{Lemma}
\newtheorem{proposition}{Proposition}
\newtheorem{cor}{Bio-Corollary}
\usepackage{graphicx}
\usepackage{picinpar,graphicx}
\usepackage{graphics}
\usepackage{multicol}

\renewcommand{\baselinestretch}{1} %for double spacing make this number 2
\bibliographystyle{plain}
%\usepackage[sort&compress]{natbib}
\setlength{\topmargin}{.5in}
\setlength{\headheight}{0in}
\setlength{\headsep}{0in}
\setlength{\textheight}{8.5in}
\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{0in}
\setlength{\evensidemargin}{0in}
\setlength{\parindent}{0.25in}
\setlength{\parskip}{0.05in}


\begin{document}
%%------------------------------------------------------------------------%%
%%-----------------------Title Page-----------------------------------%%
%%------------------------------------------------------------------------%%
\title{Your Technical Report title here}
\author{First Author$^{1}$, Second Author$^{2}$, Third Author$^{3}$, Fourth Author$^{4}$, \\ Grad Student Helper$^{5}$, Lead Advisor$^{6}$}
\date{}
\maketitle
\begin{center}
\footnotesize $^{1}$ Department, School, City, State\\
\footnotesize $^{2}$ Department, School, City, State\\
\footnotesize $^{3}$ Department, School, City, State\\
\footnotesize $^{4}$ Department, School, City, State\\
\footnotesize $^{5}$ Department, School, City, State\\
\footnotesize $^{5}$ Department, School, City, State\\
\end{center}

%%------------------------------------------------------------------------%%
%%-----------------------Abstract--------------------------------------%%
%%------------------------------------------------------------------------%%
\begin{abstract} 
{Insert Abstract here}
\end{abstract}

%%------------------------------------------------------------------------%%
%%-----------------------Introduction---------------------------------%%
%%------------------------------------------------------------------------%%


\section{Introduction}
While viewing this file, first use pdfLaTeX, then BibTeX, then pdfLaTeX again when you are running this for the first time or anytime you have added new references.  Otherwise they will come up as a question mark (?).


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Next Section}
Text here.  Be sure to cite appropriately \cite{Anderson1980,ndo,CamachoThesis,camacho,web,mtbi4,census}.  And refer to figures as Figure \ref{Fig1}, which uses
\begin{verbatim} \ref{Fig1} \end{verbatim}
to refer to this (look in the .tex file).

\subsection{Subsection title, if used}
Type more text here.

\begin{figure}[h]
\begin{centering}
\includegraphics[scale=0.4]{MTBILogo.png}
\label{Fig1}
\caption{Caption goes here}
\end{centering}
\end{figure}

\begin{figure}[h]
\begin{centering}
\includegraphics[scale=0.4]{ASUlogo.png}
\label{Fig2}
\caption{Caption goes here}
\end{centering}
\end{figure}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Discussion}
More text here.

\section*{Acknowledgments} 
We would like to thank Dr.~Carlos Castillo-Chavez, Executive Director of the Mathematical and Theoretical Biology Institute (MTBI), for giving us the opportunity to participate in this research program.  We would also like to thank Co-Executive Summer Directors Dr.~Erika T.~Camacho and Dr.~Stephen Wirkus for their efforts in planning and executing the day to day activities of MTBI. We also want to give special thanks to (specify any names of research advisors, grad helpers, etc. that you feel have helped). This research was conducted in MTBI at the Mathematical, Computational and Modeling Sciences Center (MCMSC) at Arizona State University (ASU). This project has been partially supported by grants from the National Science Foundation (NSF - Grant DMPS-1263374), the National Security Agency (NSA - Grant H98230-13-1-0261), the Office of the President of ASU, and the Office of the Provost of ASU.



%%------------------------------------------------------------------------%%
%%-----------------------BIBLIOGRAPHY---------------------------%%
%%------------------------------------------------------------------------%%

\bibliography{MTBI_bibliography}


%%------------------------------------------------------------------------%%
%%-----------------------Appendix------------------------------------%%
%%------------------------------------------------------------------------%%
\newpage
\section{Appendix}
Any Appendix goes here as Appendix A, Appendix B, etc. as separate sections.

\end{document}
